Author's response to reviews

Title: Hmw: total adiponectin rations are low in HIV-infected women receiving protease inhibitors

Authors:

Fierdoz Omar (fierdoz.omar@nhls.ac.za)
Judy King (judy.king@uct.ac.za)
Joel Dave (joel.dave@uct.ac.za)
Naomi Levitt (dinky.levitt@uct.ac.za)
Tahir S Pillay (tspillay@gmail.com)

Version: 2
Date: 5 July 2014

Author's response to reviews:

1) Line numbers have been added.
2) Details of the Ethics committee approval have been added.